Author's response to reviews

Title: Worksite interventions for preventing physical deterioration among employees in job-groups with high physical work demands: Background, design and conceptual model of FINALE

Authors:

Andreas Holtermann (aho@nrcwe.dk)
Marie B Jørgensen (mbj@nrcwe.dk)
Bibi Gram (bgram@health.sdu.dk)
Jeanette R Christensen (jrc@sport.au.dk)
Anne Faber (afh@nrcwe.dk)
Kristian Overgaard (ko@idraet.au.dk)
John Ektor-Andersen (EKTOR-ANDERSEN.JOHN@TELIA.COM)
Ole S. Mortensen (osm@nrcwe.dk)
Gisela Sjøgaard (gsjogaard@health.sdu.dk)
Karen Søgaard (ksogaard@health.sdu.dk)

Version: 4 Date: 14 February 2010

Author's response to reviews:

Ref: MS # 2765923033377200

Dear Editor,

We are very glad for your decision to publish our paper "Worksite interventions for preventing physical deterioration among employees in job-groups with high physical work demands: Background, design and conceptual model of FINALE".

We have improved the manuscript in response to the single comment by the reviewer, and inserted the suggested sentence in the revised manuscript.

On behalf of the authors,

Andreas Holtermann, PhD
National Research Centre for the Working Environment
Lersø Park Allé 105, 2100 Copenhagen Ø, Denmark
e-mail: aho@nrcwe.dk